Author’s response to reviews

Title: The changes in curriculum courses of Medicine in Brazil encouraged by Program Incentive of the Curriculum Changes

Authors:

PATRICIA MSC ALVES DE SOUZA (p.as@ig.com.br)
ANGÉLICA MARIA DR BICUDO ZEFERINO (AMBZ@FCM.UNICAMP.BR)
MARCO AURÉLIO DR DA ROS (ROS@CCS.UFSC.BR)

Version: 2 Date: 27 May 2008

Author’s response to reviews: see over
Florianópolis, 27 de maio de 2008.

Caro, MSc Natalie

As requested, the change was made in the title and summary as indicated by the instructions of the magazine. Zeferino is second author.

Carefully

Patrícia

Patricia Alves de Souza MSc
Email: p.as@ig.com.br
Tel: +55 (0) 9963 5327